University of Nebraska - Lincoln DigitalCommons@University of Nebraska - Lincoln

Faculty Publications, Department of Psychology

Psychology, Department of

7-25-2022

The role of parental health and distress in assessing children's health status

Sherrie H. Kaplan

Marilou Shaughnessy

Michelle A. Fortier

Marla Vivero-Montemayor

Sergio Gago Masague

See next page for additional authors

Follow this and additional works at: https://digitalcommons.unl.edu/psychfacpub

Part of the Psychology Commons

This Article is brought to you for free and open access by the Psychology, Department of at DigitalCommons@University of Nebraska - Lincoln. It has been accepted for inclusion in Faculty Publications, Department of Psychology by an authorized administrator of DigitalCommons@University of Nebraska - Lincoln.

Authors

Sherrie H. Kaplan, Marilou Shaughnessy, Michelle A. Fortier, Marla Vivero-Montemayor, Sergio Gago Masague, Dylan Hayes, Hal Stern, Maozhu Dai, Lauren Heim, and Zeev Kain



The role of parental health and distress in assessing children's health status

Sherrie H. Kaplan^{1,2} Marilou Shaughnessy³ · Michelle A. Fortier^{4,11,12} · Marla Vivero-Montemayor⁵ · Sergio Gago Masague⁶ · Dylan Hayes⁷ · Hal Stern⁸ · Maozhu Dai⁹ · Lauren Heim¹⁰ · Zeev Kain^{1,10,11,13}

Accepted: 28 June 2022 / Published online: 25 July 2022 $\ensuremath{\textcircled{O}}$ The Author(s) 2022

Abstract

Purpose The purpose of the study was to examine the contributions of parents' health and distress to parent's and children's assessments of children's health.

Methods We used baseline data from a longitudinal study of 364 children (ages 4–12) about to undergo surgery and their parents in a Southern California pediatric hospital. We used the 20-item child self-reported CHRIS 2.0 general health and the parallel parent-reported measure of the child's health, along with a measure of parental distress about the child's health were administered in the perioperative period. Other measures included parents' physical and mental health, quality of life, distress over their child's health, and number and extent of other health problems of the child and siblings.

Results On average, parents' reports about the child were consistently and statistically significantly higher than children's self-reports across all sub-dimensions of the CHRIS 2.0 measure. Parents' personal health was positively associated with their reports of the child's health. More distressed parents were closer to the child's self-reports, but reported poorer personal health.

Conclusion Parent–child differences in this study of young children's health were related to parental distress. Exploring the nature of the gap between parents and children in assessments of children's health could improve effective clinical management for the child and enhance family-centered pediatric care. Future studies are needed to assess the generalizability of CHRIS 2.0 to other health settings and conditions and to other racial/ethnic groups.

Keywords Children's health \cdot Children's self-reported health-related quality of life \cdot Parent proxy reporting \cdot Parental distress

Sherrie H. Kaplan skaplan@uci.edu

- ¹ Health Policy Research Institute, University of California, 100 Theory Ste 110, Irvine, CA 92697, USA
- ² Department of Medicine, University of California, Irvine, USA
- ³ Newton Wellesley Hospital, Newton, MA, USA
- ⁴ Sue & Bill Gross School of Nursing, University of California, Irvine, USA
- ⁵ Department of Psychology, University of Nebraska-Lincoln, Lincoln, NE, USA
- ⁶ Donald Bren School of Information and Computer Sciences, University of California, Irvine, USA

- ⁷ Independent Animation Consultant, New York, NY, USA
- ⁸ Department of Statistics, University of California, Irvine, USA
- ⁹ Microsoft, Bellevue, WA, USA
- ¹⁰ Department of Anesthesiology & Perioperative Care, University of California, Irvine, USA
- ¹¹ UCI Center on Stress & Health, University of California, Irvine, Irvine, USA
- ¹² Department of Pediatric Psychology, Children's Health Orange County, Orange, USA
- ¹³ Department of Pediatrics, Children's Health Orange County, Orange, USA

Plain English summary

Although many studies have found differences between parents' and children's views of the child's health, the reasons for those differences have been less well studied. Using an animated, computer-administered method to ask young children (ages 4-12) questions about their general health (called the Child Health Rating Inventories or "CHRIS") we compared their answers to those of their parents in a group of children about to undergo surgery (n = 364). We found that children gave reliable answers to CHRIS questions, and that parents rated children as healthier than their children did. We also found that parents' reports about their children were closely related to what they said about their own health. Distressed parents were closer to their children's self-reports but they also reported poorer personal health. We concluded that the CHRIS measure offers an opportunity to include the child's voice and a potentially different perspective on children's health for use by clinicians and researchers. We also concluded that exploring parent-child differences particularly for distressed parents, who may be more 'vigilant' observers of children's health but at a cost to their own health, could provide valuable insight for clinicians caring for these children and their families. Specifically, improved awareness of the perspectives and experiences of both parents and especially children could identify reasons for discrepancies between the two and therefore improve care at home and outcomes after surgery, parent-physician-child communication, and address parents' worry or distress over their child's health after surgery.

Introduction

A number of studies have documented substantial differences between reports of children's health by parents vs. by children themselves [1-7]. Although the direction of these differences varies somewhat by the disease or condition under study and the child's age and race/ethnicity, the correspondence between parent and child has been consistently low to moderate [1-7].

Although a number of hypotheses have been advanced for the lack of agreement between parent and child, [1–3, 8] the reasons for those disagreements and the implications for pediatric practice are less well empirically studied [2, 4, 5]. Some studies suggest that parents may be closer to the child's self-reports for measures involving more observable behaviors, such as those related to physical vs. mental health [4]. Others have suggested that parent-child differences may be narrower for younger children [5, 9]. Still other studies suggest that parents may use their own health as a reference when reporting their child's health [5].

Some subgroups of parents (e.g., those with children who have chronic conditions) who regularly or closely monitor their children's health may be closer to the child's self-perceptions, thus reducing the gap between parents' and children's reports [1–7]. Such monitoring or "parental vigilance," however, may come at a price. Some research suggests that among children with chronic diseases, closer parental monitoring, while improving treatment outcomes, may lead to greater parental distress and poorer personal health [10, 11].

A first step in identifying contributors to parent-child differences in assessments of children's health is to avoid the potential bias introduced by proxy reporting, particularly for young children [12, 13]. We have developed and previously tested a measure of health-related quality of life specifically designed for independent self-reporting by young children (ages 4–12) [14, 15]. This 20-item measure, the child health rating inventories (CHRIS) [14, 15], uses computer-administered animation to display and record item content and responses, in eight health-related quality of life dimensions scored in real time.

The purpose of the current study was to examine the contributions of parental health and distress to parents' and children's self-assessments of the child's health. In a population of children about to undergo outpatient surgery or minor surgical procedures, we used the CHRIS 2.0 to measure the children's self-reported health along with parents' reports of their own health and their distress over the participating child's health. We also included characteristics previously observed to contribute to parent-child differences, including the child's age, race/ethnicity, gender, and illnesses, as well as parents' age, education, race/ethnicity, and gender, family size, and extent of illness among the child's siblings [2-6, 16–18]. We hypothesized (1) that, based on the literature and our prior work, there would be discrepancies between parents' and children's reports in this population and (2) that, adjusted for the characteristics listed above, parents who reported greater distress over the child's health would report poorer health for the child and for themselves as care givers. We report here the results of that study.

Methods

Study design

This is a cross-sectional study reporting baseline data for 364 parent–child dyads participating in a longitudinal cohort study of children ages 4 to 12, undergoing outpatient surgery, described in detail elsewhere [14].

Site and sample

The study was conducted at a 232-bed pediatric hospital serving a largely poor and minority population in Southern California. Children and their parents were sampled from outpatient surgical logs screened weekly for eligibility from September 1, 2015 through August 30, 2017. Eligibility criteria determined from the child's medical record and confirmed by parents where relevant were children ages (4–12 years); surgical procedure was either outpatient but not an emergency (defined as urgent surgery not scheduled in advance) or associated with a cancer diagnosis; pre-operative status of the American society of anesthesiologists (ASA) classification of I (a normal healthy patient) or II (a patient with mild systemic disease) (see: https://www.asahq.org/standards-and-guidelines/asa-physical-status-classifica

tion-system); no developmental delays or special needs; and both parent and child spoke English or Spanish. Eligible families were offered participation in the study, consented, enrolled, and completed the CHRIS 2.0 child and parent measures 3–7 days prior to surgery. Participating families were emailed or texted electronic links to complete the survey online. Reminders were sent by phone or text to those who had not responded 3 days prior to surgery. The analytic sample included 364 parent–child dyads with complete baseline data for the study variables included in this study.

Study measures

hildren's Health Rating

The CHRIS measures have been described in detail elsewhere [14, 15, 18, 19]. In brief, the updated CHRIS 2.0 measure for children was administered by computer. The 20 items representing eight dimensions of health and response options were "read" to the child by a previously recorded narrator. A frozen frame of each animated five-level Likerttype response option appeared at the bottom of the screen for each item, and the child touched the response that most closely matched their current health state. Each response was recorded and stored in an encrypted database via a secure platform. We have provided sample item content with a static image below.



"How much does not feeling well get in the way when you want to play hard? (Like run fast, bike hard, play tag)"



A more complete display of item content, with detailed psychometric analyses appears elsewhere [14]. Based on results from factor analyses [14], a physical health composite was created from 12 items representing physical, role, cognitive, and social function and a mental health composite of 8 items representing mental health, pain, energy, and overall quality of life was created. We also created an overall composite that included all 20 items. CHRIS 2.0 parent questionnaires with parallel content were completed online using the survey platform Qualtrics [20]. All scales have been transformed to range from 0 to 100.

We used 4 items from the CHQ-Parent Form 28 (CHQ-PF28) [21] to assess parental distress, specifically "During the past four weeks, how often did your child's physical health cause you: emotional suffering or worry?"; limit the amount of time you had for your own "personal needs?" and the same 2 items were repeated for the child's emotional health. The responses were rated on a five-point Likert scale from "all of the time" to "none of the time." Parents' reports about their own physical function, role function, social function, cognitive function, and energy/vitality were measured using items paralleling those from the child-reported CHRIS 2.0, rated on a five-point Likert scale and transformed to range from 0 to 100. Parents' mental health was assessed using the 5-item Mental Health Index of the Short Form 36 (SF-36) [22]. Overall quality of life, reported by the parent for the child and for their own quality of life, was measured by the 8-item quality of life rating scale (QLRS), with individual items representing each of eight areas of life (work/ school life, family life, friendships, daily routine, physical health, mental health, personal/"playtime," or fun and general life enjoyment), also rated on a 5-point Likert scale from "excellent" to "poor" [23]. Parents were also asked to report the number of the child's siblings, the number and extent of sibling illnesses, and the participating child's illnesses (measured as a composite of the presence and severity of 12 chronic conditions-diabetes, thyroid disease, growth problems, obesity, underweight, asthma, seizures, allergies, bowel disease, ADHD, developmental delays, and any other illnesses).

Statistical approach

Data were analyzed using R [24] and SAS software [25]. We used a variety of summary measures (means, standard deviations, kurtosis, skew) to assess the distributional shape of responses to individual items and scales. We used Cronbach's alpha [26] to assess internal consistency reliability and factor analysis with varimax rotation to replicate hypothesized scale structure for the CHRIS 2.0 measures (as described elsewhere) [14]. Paired *t* tests were used to assess differences in the mean responses of parents and children on the CHRIS 2.0 measures. One-way random effects intraclass

correlation coefficients provided a quantitative measure of the degree of agreement between parent- and child-reported CHRIS 2.0 measures. Parents' reports about their own health were compared with their reports about their children using Pearson product–moment correlation coefficients. The contribution of parent, child, and family characteristics to differences between parents' and children's reports about the child's health for the physical, mental, and overall composite measures were estimated using separate parsimonious linear regression models for each composite.

Mean differences between parent- and child-reported CHRIS 2.0 measures for most, moderately, and least distressed parents were compared using separate analyses of covariance, adjusting for parental age, education, race, number of siblings in the family, sibling illnesses, child's age and number of health problems, and types of surgery. Comparisons of the parents' physical health, mental health and overall quality of life for those parents who were in the highest quartile vs. remaining quartiles of self-reported

Table 1 Characteristics of the child and parent cohort samples^a

	Child sample $(n=364)$	Parents sample $(n=364)$
Demographic characteristics		
Age (mean yrs)	7.5 [2.6]	36.7 [7.1]
Female (%)	40.7	86.6
Education (mean yrs)	2.3 [2.7]	13.3 [3.4]
Race/ethnicity (%) ^b		
Non-Hispanic white	24.1	28.2
Hispanic	68.1	63.3
Asian	6.0	6.2
African American	0.9	0.3
Children's health problems		
Number of health problems ^b (mean)		2.0 [2.7]
Family characteristics		
Number of children in family ^b	-	3.1 [1.6]
Other children's illness ^c		0.7 [2.0]
Surgical type (%)		
ENT^{d}	51.3	
Urology	16.4	
General	11.9	
Plastic	5.9	
Other	14.5	

^aTable entries are means with standard deviations in parentheses or percentages as indicated; report by children and parents of 364 dyads from the pre-operative sample (and 146 dyads from the post-operative sample)

^bSelf-reported by parents

^cTotal number of chronic illnesses among participating child's siblings (e.g., diabetes, asthma, weight problems, and allergies)

^dEar, nose, and throat

distress about the child's health, were performed using analysis of variance, adjusted for parental age, education, race, number of children in the family, extent of the child's siblings' illnesses, the child's age, number of the child's health problems, and type of surgery. Mean differences between child- and parent-reported CHRIS 2.0 measures by age of the child were compared using separate analyses of variance for each scale and composite measure.

Missing data

The overall rate of missing data was < 10%, with < 5% of CHRIS 2.0 items missing. Given the limited amount of missing data, we carried out study analyses using only cases with complete data.

IRB approval

The study was approved as minimal risk by the Institutional Review Board at the study site.

Results

Characteristics of the 364 parent–child dyads included in this study appear in Table 1. The parents were moderately well-educated, approximately two-thirds were Hispanic, the majority were female (86.6%) and in their mid-30's. Children were on average 7.5 years of age, with somewhat fewer females than males. Roughly 14% of the children in the study were under age 5. Including the participating child, there were approximately three children per family, with few 0.7 of the twelve chronic illnesses measured among siblings and roughly two of these twelve illnesses per participating child.

Internal consistency reliability coefficients for the CHRIS 2.0 subscales and composite measures for children and parents in this sample have been reported elsewhere [14]. Parents' reports about their children's health were consistently and significantly higher than children's self-reports for each of the CHRIS 2.0 subscales and composite measures with the exception of overall quality of life (see Table 2). Children reported better CHRIS 2.0 overall quality of life scores than the parents' reports about the child. The magnitude of these differences averaged approximately 60% of a standard deviation across CHRIS 2.0 measures. The level of agreement

CHRIS 2.0 measures Mean difference (±95% CI)^b ESc ICC^d Reporter Child Parent General health measures Physical function 61.9 [25.1] 80.1 [30.4] -18.2^{***} (-21.8, -14.6)0.65 .09 - 19.6*** Role function 67.1 [24.7] 86.7 [24.8] (-23.0, -16.3)0.79 - .01 - 15.1*** Social function 87.1 [23.9] .07 72.0 [25.9] (-18.4, -11.8)0.61 Cognitive function - 20.3*** (-23.8, -16.8)0.80 - .05 67.3 [26.8] 87.6 [24.2] - 15.8*** Energy 68.6 [22.7] 84.4 [27.1] (-19.0, -12.6)0.63 .13 - 6.2*** Paine 74.2 [28.5] 80.5 [25.0] (-9.3, -3.2)0.23 .36 - 6.1*** (-8.7, -3.5)Mental health 73.6 [25.3] 79.7 [14.0] 0.31 .22 Overall quality of life 82.3 [20.6] 78.2 [20.2] 4.2*** (1.8, 6.5)0.21 .36 - 18.8*** (-21.7, -16.0)0.86 .01 Physical health composite 66.2 [20.6] 85.0 [23.3] - 6.0*** Mental health composite (-7.7, -4.2)0.38 74.0 [18.2] 79.9 [13.5] .39 - 12.7*** Overall composite 69.3 [17.5] 82.0 [15.5] (-14.6, -10.8)0.77 .20

p < 0.05, p < 0.01, p < 0.01

^aTable entries are means with standard deviations in parentheses; scale scores have been transformed to range from 0 to 100 with high scores indicating better health

^bMean differences are entered as child-reported scores minus parent-reported scores for each measure with 95% confidence intervals for mean differences in parentheses; statistical significance is based on F-statistic using one-way ANOVA, adjusting for parental age, education, race, number of the child's siblings, extent of sibling illness, child's age, and number of the child's health problems

^cEffect size (mean difference expressed as a proportion of the pooled standard deviation)

^dIntra-class correlation coefficients for agreement between child and parents scores on each measure (estimated using one-way random effects models and method of moments); scores < .40 are considered poor agreement³⁹

ePain was transformed such that higher score indicates less pain

Table 2 Mean differencesbetween child- and parent-reported CHRIS 2.0 measures $(n = 364)^a$

Table 3 Mean differences between child- and parentreported CHRIS2.0 measures for parents with higher/lower levels of distress $(n = 352)^{a}$

Table 4Personal health ofmost vs. less distressed parents

 $(n=352)^{a}$

CHRIS2.0 measures ^b	Least distressed ^c $(n=113)$	Moderately distressed ^d $(n=159)$	Most distressed ^e (n=80)	Adjusted <i>p</i> -value ^f
Physical function	- 27.4	- 18.6	- 9.1	0.009
Role function	- 26.4	- 18.4	- 13.5	0.009
Social function	- 21.9	- 12.7	- 10.9	0.011
Cognitive function	- 21.5	- 21.7	- 17.3	0.555
Energy	- 22.9	- 15.4	- 8.7	0.004
Pain	- 13.7	- 4.5	- 1.6	0.042
Mental health	- 4.6	- 5.8	- 9.8	0.793
Overall quality of life	- 0.2	3.3	9.7	0.194
Physical health composite	- 25.4	- 18.4	- 12.5	0.005
Mental health composite	- 8.6	- 5.5	- 4.3	0.105
Overall composite	- 17.5	- 12.3	- 8.3	0.002

^aMeasured using 4-item scale on the impact of the child's physical and emotional health on parental wellbeing from the CHQ- $PF28^{30}$

^bCorresponding measures of each CHRIS2.0 construct reported by child and parent

^cReported as the lowest (least distressed) quartile of scores on the parental distress scale

^dReported as the intra-quartile of scores (moderately distressed) on the parental distress scale

^eReported as the highest (most distressed) quartile scores on the parental distress scale

^f*P*-values based on ANOVA adjusting for parental age, education, race, number of siblings in the family, sibling illnesses, participating child's age, and total number of participating child's health problems

Health measures	Parental distress ^b		Mean difference (±95% CI) ^e
	Most distressed ^c $(n=80)$	Less distressed ^d $(n=272)$	
Physical health ^f	82.8 [3.5]	87.3 [2.7]	- 4.5 [§] (- 10.6, 1.5)
Mental health ^g	61.7 [2.4]	75.9 [1.9]	- 14.2*** (- 18.3, - 10.0)
Quality of life ^h	63.3 [3.2]	74.3 [2.5]	- 11.0**** (- 16.6, - 5.3)

§*p*=.141

***p < .001

^aTable entries are adjusted means with standard errors in parentheses based on analysis of covariance adjusted for parental age, education, race, number of siblings in the family, sibling illnesses, child's age, and number of health problems

^bParental distress was measured using 4 items from the CHQ-Parent Form 28³⁰ related to impact of the child's health on the parent's emotional well-being and time for personal needs

^cParents scoring in the highest quartile of parental distress

^dParents scoring in the 3 quartiles of parental distress other than the highest, combined

^eMean difference calculated as less minus more distressed parent scores for each health measure, with 95% confidence intervals around mean differences

^fMeasured as a composite of the 7-item CHRIS2.0 parent physical function scale

^gMeasured as a composite of the 10-item CHRIS2.0 parent mental health scale

^hMeasured using the 8-item QLRS³⁵

between parents and children, as indicated by the intraclass correlation coefficients, was < 0.40, a level considered as "poor" agreement [27] (see Table 2). Mental health measures had generally higher levels of parent–child agreement, compared with physical health measures. The magnitude of the differences between child- and parent-reported CHRIS

2.0 measures did not differ by age of the child (see Appendix A).

Parents' reports about their children's health were significantly correlated with their reports about their own physical, mental, overall health, and quality of life (see columns 1–3, 5, Appendix B). We also observed significant correlations between parents' reports about their children's health and level of distress about their child's health, r > 0.30, p < 0.01 (see column 4, Appendix B).

Compared to moderately and least distressed parents, the most distressed parents gave consistently lower ratings of their children's health, across CHRIS 2.0 measures (see Table 3). They were thus closer to their children's selfreports. These data have been adjusted for parental age, years of education, race/ethnicity, distress, quality of life, number of the child's siblings, sibling illness, child's age, and number of health problems. Of these variables in linear regression models assessing parent–child differences, only parental distress was significantly associated with differences in child–parent reports (data not shown).

Parents scoring in the highest quartile of parental distress about their child's health also gave substantially and statistically significantly lower ratings of their own mental health and quality of life compared to those scoring in the remaining quartiles (see Table 4). Compared to distressed parents, those with greater distress about their child's health also reported poorer personal physical health, although this difference did not reach statistical significance. These analyses were also adjusted for parental age, education, race, number of children in the family, extent of sibling's illness, the participating child's age, and number of health problems.

Discussion

Consistent with a substantial body of existing literature [5, 6, 16, 28], we found significant differences between parents' reports about their children's health and children's self-reports using the CHRIS 2.0 measure. The perspective parents in our study adopted, as a proxy reporter or from their own view of the child's health, was not explicitly addressed. As noted by Snow AL et al. [12], the definition of a 'proxy' is data "collected from someone who speaks for a patient who cannot, will not, or is unavailable to speak for him or herself" and is therefore attempting to provide information on the same construct as would the patient if s/he were able to provide that information themselves. It is not clear whether the poor agreement between parent and child reports of the child's health across multiple studies suggests that parents were acting as proxy or other rater reporters about the child's health. Our instructions to parents were only to ask them to rate their child's health with the implicit assumption that they were acting as proxy reporters. It would be expected therefore that there should be similarities between the two respondents if both perspectives are obtained. But as Pickard AS, et al., 2005 [29] has suggested, the instructions given to the proxy, including to "assess the patient as they think the patient would respond" vs. provide their own perspective

on the patients health (i.e., same construct, different perspective), would perhaps provide a valuable clarifying context for proxy reporting in future research.

Also consistent with other research we found that parents' assessments of their children's health were positively correlated with reports of their own health [5, 30–32]. We found that parents who were in greater distress about their child's health also reported poorer personal health, but were closer to their child's self-reports.

That parents and children differ in assessments of the child's health is supported by a number of studies and systematic reviews that span a broad spectrum of diseases, populations, and clinical settings [1–7]. Most of these studies used well-tested measures of young children's health-related quality of life, such as the Pediatric Quality of Life Inventory (PedsQL) [28], the Functional Disability Inventory (FDI) [33], KIDSCREEN [34], and the Child Health Questionnaire (CHQ) [35], that require interviewer assistance or parent report for children who cannot read. Unique to this study is the use of the CHRIS measure for children as young as 4 years of age to self-report their health-related quality of life. Such self-reporting, independent of parents or interviewers, can avoid the potential proxy-reporting bias observed for children and adults [13, 16, 29, 36].

Parents in our study over-estimated their child's selfreported health. This finding is consistent with other studies of parent-child differences among "healthy" children [1, 5]. Although children in our sample were about to undergo surgery, the majority did not have chronic diseases and surgeries or procedures were outpatient or not serious, making them more similar to a "healthy" child sample.

We also observed fewer parent-child differences for mental vs. physical health measures. This finding was not consistent with the hypothesis that parents were better at approximating the child's self-reports for more "observable" behaviors [4, 5, 9]. It may be that the animation of content afforded by the CHRIS measure more effectively communicated the feelings and emotions related to the mental health items or that parents were more attuned to the emotional state of their children in the perioperative period. One feasibility study in the UK of animation vs. text presented by tablet or on paper assessing children's self-reported health utilities, administered to children ages 4-14 in school and hospital settings, found that children preferred animation over either text presentation and even the youngest age group were able to understand the animated questions [37]. More research is needed to explore the impact of animation on parent-child differences in health-related quality of life. We did not find evidence of fewer or greater differences between children and parents by the age of the child.

The finding that parents report about their child were positively correlated with their personal health supported the "self-referencing" hypothesis proposed by some researchers [5, 36]. That is, in reporting about the health of another, proxies use their own health as the referent. In comparisons among adults, for example, proxies appear to report lower health-related quality of life scores (or under-estimate health) for elderly patients than the elderly themselves [36]. In contrast, as noted above, parents of healthy children tended to over-estimate the child's health, especially for young children [6].

Although cited as a need for further research [3, 5, 6], few studies have empirically examined the factors contributing to parent-child differences in reporting health-related quality of life. We found that, adjusted for demographic characteristics of children and parents and the number and severity of health problems of the participating children and their siblings, parental distress was the principal contributor to parent-child differences in CHRIS 2.0 scores. Further, compared to less distressed parents, those reporting greater distress over their child's health were closer to the child's self-reported CHRIS 2.0 scores. This finding is consistent with other studies among parents of children with chronic diseases [11, 30, 38, 39].

A few studies of parent-child differences in reports of children's health have found similar results, namely more distressed parents report lower scores for their child's health [10, 32, 40]. Our study was not designed to explore the causal direction of this relationship. The illness of a child could certainly cause some parents greater distress than others or the attentiveness required to attain a better understanding of the child's perception of their health could cause greater parental worry or concern and limit their attention to their own needs. Distressed parents in our study also reported poorer personal health, raising questions and concerns about the toll "vigilant parenting" may be taking on parents [38]. This toll could be greater for parents of children with chronic diseases [10, 38]. Understanding the extent of the gap between parent and child perspectives on the child's health, the relationship of parental distress to that gap and the effects of parental distress on the overall health of parents as caregivers could have important implications for effective and family-centered pediatric care [40].

As the pace and scope of interventions to improve pediatric care has accelerated, the need to evaluate their effectiveness, not using only clinical and biomarkers of health, but also their impact on children's health-related quality of life has been increasingly recognized [2]. The CHRIS 2.0 measure is a reliable and valid health-related quality of life measure [14, 15] and allows young children to report their health independent of adults in busy clinical settings.

Despite this independent reporting, our study highlighted not only the persistence of parent-child differences in perceptions of children's health, but also the importance of parental distress related to both the magnitude and direction of these differences. If, as recommended by some, differences between parent and child are not as biased or inaccurate in reporting by one or the other, but rather as a "quantifier of the gap" [12], including both could provide a more developed or comprehensive picture of the child's health. Exploring the nature of that gap during office visits, including parallel assessments of parental distress over the child's health, could have important implications for both effective clinical care for the child [2-4, 16], for family-physician communication, for family-centered pediatric care [41, 42], and for parents' health and well-being. Integration of measures of children's health-related quality of life, such as the CHRIS 2.0, and parent's health as additional "vital signs" in routine pediatric practice [43] could provide valuable and unique information for clinicians and researchers evaluating and tailoring treatment regimens and improving pediatric practice.

Limitations

This study was conducted at a single pediatric academic healthcare institution. Whether the findings of this study replicate in other healthcare settings requires additional study. Also, although the study sample was highly diverse, some racial/ethnic groups, including African Americans, were underrepresented. Differences in parent-child reports of children's health in other such racial and ethnic groups will require additional study. We also studied children about to undergo elective surgical procedures. Although our prior work suggests the current findings replicate among children with chronic conditions [15], CHRIS 2.0 will require additional testing in populations of children with other conditions and among "healthy" children in the included age range. Finally, in a recent study [14], we tested the CHRIS 2.0 among children as young as 4 years of age and found the reliability of their reports to parallel those for older age groups. In this study we also observed that differences in parent-child reports for the youngest children also paralleled the findings for older age groups. We do not know whether reliability findings and parent-child differences for even younger children would produce findings observed for this study. Finally, we did not directly assess whether and to what extent parents assisted children to complete the CHRIS2.0 measure at home. However, to the extent that parents influenced the child's reports, more alignment of responses would have been expected, which we did not find.

Conclusion

The differences between parent and child in reports about the child's health are striking and suggest an opportunity for clinicians to explore the nature of those differences. The CHRIS 2.0 offers an opportunity to include the child's voice in the direct reporting of their own health-related quality of life without assistance from adults. It can produce reliable and valid results even among young children. In light of the finding that more distressed parents provided assessments that were closer to the child's self-reports, exploring the nature and potential cost of parental vigilance on the parents' own health could also provide valuable insight for the clinicians caring for these children and their families. Further studies are needed to assess the generalizability of CHRIS 2.0 in other health settings and conditions and to other racial/ethnic groups.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11136-022-03186-z.

Author contributions SHK conceptualized and designed the CHRIS2.0 measure, conceptualized and designed the study, formulated and oversaw data analysis, drafted the initial manuscript, and reviewed and revised the manuscript. MF and ZK conceptualized and designed the study, designed data collection instruments, including measures other than the CHRIS, coordinated and supervised data collection, and critically reviewed the manuscript for intellectual content. MS assisted with the development of the CHRIS modules, oversaw field operations, including data collection, recommended appropriate analyses, and reviewed and revised the manuscript. SGM designed and maintained the integrated database for study analyses, oversaw field operations, ensured protocol fidelity, and critically reviewed and revised the manuscript. M and MVM developed the field operation protocol, implemented the study design, coordinated and supervised data collection, and reviewed and revised the manuscript. DH designed and implemented the CHRIS animations, assisted with the development of data collection protocols ensuring fidelity of multiple data collection platforms, and reviewed and revised the manuscript. HS and MD conducted data analysis and reviewed and revised the manuscript for intellectual content and statistical appropriateness. LH coordinated and assisted with data interpretation and reviewed and revised the manuscript. All authors approved the final manuscript as submitted and agree to be accountable for all aspects of the work.

Funding This study was supported by the Patient-Centered Outcomes Research Institute [ME-1306–01715] and the University of California, Irvine Health Policy Research Institute. The funder had no role in the study design; in the collection, analysis, and interpretation of data; in the writing of the report; and in the decision to submit the article for publication.

Declarations

Conflict of interest The authors have no conflicts of interest relevant to this article to disclose.

Ethics approval The study was approved as minimal risk by the Institutional Review Board at the study site.

Informed consent Informed consent was obtained from all individual participants included in the study. Written informed consent was obtained from the parents.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source,

provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Galloway, H., & Newman, E. (2017). Is there a difference between child self-ratings and parent proxy-ratings of the quality of life of children with a diagnosis of attention-deficit hyperactivity disorder (ADHD)? A systematic review of the literature. *ADHD Attention Deficit and Hyperactivity Disorders*, 9(1), 11–29. https://doi. org/10.1007/s12402-016-0210-9
- Upton, P., Lawford, J., & Eiser, C. (2008). Parent-child agreement across child health-related quality of life instruments: A review of the literature. *Quality of Life Research*, 17(6), 895–913. https:// doi.org/10.1007/s11136-008-9350-5
- Lee, Y. C., Yang, H. J., Lee, W. T., & Teng, M. J. (2019). Do parents and children agree on rating a child's HRQOL? A systematic review and Meta-analysis of comparisons between children with attention deficit hyperactivity disorder and children with typical development using the PedsQLTM. *Disability and Rehabilitation*, 41(3), 265–275. https://doi.org/10.1080/09638288.2017.1391338
- Eiser, C., & Morse, R. (2001). Can parents rate their child's health-related quality of life? Results of a systematic review. *Quality of Life Research*, 10(4), 347–357. https://doi.org/10. 1023/a:1012253723272
- Cremeens, J., Eiser, C., & Blades, M. (2006). Factors influencing agreement between child self-report and parent proxy-reports on the pediatric quality of life inventory 4.0 (PedsQL) generic core scales. *Health Qual Life Outcomes*, 4, 58. https://doi.org/10.1186/ 1477-7525-4-58
- Yi-Frazier, J. P., Hilliard, M. E., Fino, N. F., et al. (2016). Whose quality of life is it anyway? Discrepancies between youth and parent health-related quality of life ratings in type 1 and type 2 diabetes. *Quality of Life Research*, 25(5), 1113–1121. https://doi. org/10.1007/s11136-015-1158-5
- Varni, J. W., Limbers, C. A., & Burwinkle, T. M. (2007). Parent proxy-report of their children's health-related quality of life: an analysis of 13,878 parents' reliability and validity across age subgroups using the PedsQL 4.0 generic core scales. *Health Qual Life Outcomes.*, 5, 2. https://doi.org/10.1186/1477-7525-5-2
- Husbands, S., Mitchell, P. M., & Coast, J. (2020). A systematic review of the use and quality of qualitative methods in concept elicitation for measures with children and young people. *Patient*, 13(3), 257–288. https://doi.org/10.1007/s40271-020-00414-x
- Rajmil, L., López, A. R., López-Aguilà, S., & Alonso, J. (2013). Parent-child agreement on health-related quality of life (HRQOL): A longitudinal study. *Health and Quality of Life Outcomes*, 11, 101. https://doi.org/10.1186/1477-7525-11-101
- DeMore, M., Adams, C., Wilson, N., & Hogan, M. B. (2005). Parenting stress, difficult child behavior, and use of routines in relation to adherence in pediatric asthma. *Children's Health Care.*, 34(4), 245–259. https://doi.org/10.1207/s15326888c hc3404_1
- 11. Driscoll, C. F. B., Buscemi, J., & Holmbeck, G. N. (2018). Parental distress and stress in association with health-related quality of life in youth with spina bifida: A longitudinal study. *Journal*

of Developmental and Behavioral Pediatrics, 39(9), 744–753. https://doi.org/10.1097/DBP.00000000000603

- Snow, A. L., Cook, K. F., Lin, P. S., Morgan, R. O., & Magaziner, J. (2005). Proxies and other external raters: methodological considerations. *Health Services Research.*, 40, 1676–1693. https:// doi.org/10.1111/j.1475-6773.2005.00447.x
- Garbarski, D. (2014). Comparing self and maternal reports of adolescents' general health status: Do self and proxy reports differ in their relationships with covariates? *Quality of Life Research*, 23(7), 1953–1965. https://doi.org/10.1007/s11136-014-0649-0
- Kaplan, S. H., Fortier, M. A., Shaughnessy, M., et al. (2021). Development and initial validation of self-report measures of general health, preoperative anxiety, and postoperative pain in young children using computer-administered animation. *Paediatric Anaesthesia*, 31(2), 150–159. https://doi.org/10.1111/pan.14068
- Kaplan, S. H., Barlow, S., Spetter, D., Sullivan, L., Khan, A., & Grand, R. (1995). Assessing functional status and health-related quality of life among school-aged children: reliability and validity of a new self-reported measure. *Qual Life Res.*, 4(5), 444.
- Kaminsky, O., Fortier, M., Jenkins, B., Stevenson, R. S., Gold, J. I., Zuk, J., Golianu, B., Kaplan, S. H., & Kain, Z. N. (2019). Children and their parents' assessment of postoperative surgical pain: agree or disagree? *Psychology Faculty Articles and Research*. https://doi.org/10.1016/j.ijporl.2019.04.005
- Rajmil, L., Fernández, E., Gispert, R., et al. (1999). Influence of proxy respondents in children's health interview surveys. *Journal* of Epidemiology and Community Health, 53(1), 38–42. https://doi. org/10.1136/jech.53.1.38
- Parsons, S. K., Barlow, S. E., Levy, S. L., Supran, S. E., & Kaplan, S. H. (1999). Health-related quality of life in pediatric bone marrow transplant survivors: According to whom? *International Journal of Cancer Supplement*, *12*, 46–51. https://doi.org/10.1002/ (sici)1097-0215(1999)83:12+%3c46::aid-ijc9%3e3.0.co;2-c
- Parsons, S. K., Shih, M. C., Mayer, D. K., et al. (2005). Preliminary psychometric evaluation of the child health ratings inventory (CHRIs) and disease-specific impairment inventory-hematopoietic stem cell transplantation (DSII-HSCT) in parents and children. *Quality of Life Research*, 14(6), 1613–1625. https://doi.org/10.1007/s11136-005-1004-2
- Qualtrics International Inc. Qualtrics. Published 2019. https:// www.qualtrics.com/
- Raat, H., Botterweck, A. M., Landgraf, J. M., Hoogeveen, W. C., & Essink-Bot, M. L. (2005). Reliability and validity of the short form of the child health questionnaire for parents (CHQ-PF28) in large random school based and general population samples. *Journal of Epidemiology and Community Health*, 59(1), 75–82. https://doi.org/10.1136/jech.2003.012914
- Ware, J. E., & Sherbourne, C. D. (1992). The MOS 36-item shortform health survey (SF-36). I. Conceptual framework and item selection. *Medical Care*, 30(6), 473–483.
- Greenfield, S., Kaplan, S. H., Silliman, R. A., et al. (1994). The uses of outcomes research for medical effectiveness, quality of care, and reimbursement in type II diabetes. *Diabetes Care*, *17*(Suppl 1), 32–39.
- 24. R Core Team. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. Published online 2020. https://www.R-project.org
- 25. Osborne, J., & Banjanovic, E. (2016). *Exploratory factor analysis with SAS* (1st ed.). SAS Institute Inc.
- Yurdugül, H. (2008). Minimum sample size for cronbach's coefficient alpha: A monte-carlo study. *Hacettepe Üniversitesi Eğitim Fakültesi Dergisi.*, 35(35), 1–9.
- Koo, T. K., & Li, M. Y. (2016). A guideline of selecting and reporting intraclass correlation coefficients for reliability research. *Journal of Chiropractic Medicine*, 15(2), 155–163. https://doi.org/ 10.1016/j.jcm.2016.02.012

- Varni, J. W., Seid, M., & Kurtin, P. S. (2001). PedsQL 4.0: reliability and validity of the pediatric quality of life inventory version 4.0 generic core scales in healthy and patient populations. *Medical Care*, 39(8), 800–812. https://doi.org/10.1097/00005650-200108000-00006
- Pickard, A. S., & Knight, S. J. (2005). Proxy evaluation of healthrelated quality of life. *Medical Care*, 43(5), 493–499.
- Janicke, D. M., Marciel, K. K., Ingerski, L. M., et al. (2007). Impact of psychosocial factors on quality of life in overweight youth. *Obesity*, *15*(7), 1799–1807. https://doi.org/10.1038/oby.2007.214
- Vetter, T., Bridgewater, C. L., & Mcgwin, G. (2012). An observational study of patient versus parental perceptions of health-related quality of life in children and adolescents with a chronic pain condition: Who should the clinician believe? *Health and Quality of Life Outcomes*. https://doi.org/10.1186/1477-7525-10-85
- 32. Galloway, H., Newman, E., Miller, N., & Yuill, C. (2019). Does parent stress predict the quality of life of children with a diagnosis of ADHD? A comparison of parent and child perspectives. *Journal of Attention Disorders*, 23(5), 435–450. https://doi.org/ 10.1177/1087054716647479
- Walker, L. S., & Greene, J. W. (1991). The functional disability inventory: Measuring a neglected dimension of child health status. *Journal of Pediatric Psychology*, 16(1), 39–58. https://doi.org/10. 1093/jpepsy/16.1.39
- Ravens-Sieberer, U., Gosch, A., Abel, T., et al. (2001). Quality of life in children and adolescents: A European public health perspective. *Sozial- und Praventivmedizin*, 46(5), 294–302. https:// doi.org/10.1007/BF01321080
- 35. Landgraf, J., Abetz, L., & Ware, J. (1996). *The CHQ User's Manual*. The Health Institute, New England Medical Center.
- Li, M., Harris, I., & Lu, Z. K. (2015). Differences in proxyreported and patient-reported outcomes: Assessing health and functional status among medicare beneficiaries. *BMC Medical Research Methodology*, *15*, 62. https://doi.org/10.1186/ s12874-015-0053-7
- Guerriero, C., Jaume, N. A., Diaz-Ordaz, K., et al. (2020). Using animation to self-report health: A randomized experiment with children. *Patient*, *13*(2), 175–188. https://doi.org/10.1007/ s40271-019-00392-9
- Cousino, M. K., & Hazen, R. A. (2013). Parenting stress among caregivers of children with chronic illness: A systematic review. *Journal of Pediatric Psychology.*, 38(8), 809–828. https://doi.org/ 10.1093/jpepsy/jst049
- Grimaldi Capitello, T., Fiorilli, C., Placidi, S., Vallone, R., Drago, F., & Gentile, S. (2016). What factors influence parents' perception of the quality of life of children and adolescents with neurocardiogenic syncope? *Health and Quality of Life Outcomes*, 14, 79. https://doi.org/10.1186/s12955-016-0476-9
- Moreira, H., Carona, C., Silva, N., Frontini, R., Bullinger, M., & Canavarro, M. C. (2013). Psychological and quality of life outcomes in pediatric populations: A parent-child perspective. *Journal of Pediatrics*, *163*(5), 1471–1478. https://doi.org/10.1016/j. jpeds.2013.06.028
- Chorney, J. M., & Kain, Z. N. (2010). Family-centered pediatric perioperative care. *Anesthesiology*, *112*(3), 751–755. https://doi. org/10.1097/ALN.0b013e3181cb5ade
- Claus, L. E., Links, A. R., Amos, J., et al. (2021). Parent experience of communication about children's surgery: A qualitative analysis. *Pediatr Quality Safety*, 6(3), e403. https://doi.org/10. 1097/pq9.00000000000403
- Bele, S., Chugh, A., Mohamed, B., Teela, L., Haverman, L., & Santana, M. J. (2020). Patient-reported outcome measures in routine pediatric clinical care: a systematic review. *Frontiers in Pediatrics*, 8, 364. https://doi.org/10.3389/fped.2020.00364

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.